Author’s response to reviews

Title: Incomplete ileus and hemafecia as the presenting features of multi-organ involved primary systemic AL amyloidosis: a rare case report

Authors:

Li Tian (fieldpower927@163.com)
Anliu Tang (13875893450@163.com)
Xian Zhang (1192463160@qq.com)
Zhen Mei (897449572@qq.com)
Fen Liu (liufencsu@163.com)
Jingbo Li (355704560@qq.com)
Xiayu Li (lixiayu@163.com)
Feiyan Ai (522688650@qq.com)
Xiaoyan Wang (wxy20011@163.com)
Shourong Shen (ssrxy3yy@163.com)

Version: 1 Date: 10 Nov 2016

Author’s response to reviews:

For the Reviewer #2:

Reply: 1) The information about incidence and prevalence of amyloidosis has been added in the manuscript. In this paper, it showed the incidence of amyloidosis is not well reported actually, but probably falls between 5 and 13 per million per year. Prevalence data are also scare, a study in UK reported about 20 per million local people.

2) Because it abruptly happened GI bleeding in this patient, and the internal medicine could not stop the hemorrhage and the low blood pressure. We had no other choice but the surgery to save his life. After the operation, we recall all of these process, we consider the hemorrhage in GI tract also caused by the primary disease, namely primary systemic AL amyloidosis

3) The patient was treated with MPT chemotherapy regimen per 1 or 2 months till now. During our following up process, his disease is well controlled.
Reference: